Author’s response to reviews

Title: Lymph nodal Fine Needle Cytology in the Staging and Follow-up of Cutaneous Lymphomas

Authors:

Elena Vigliar (elena@vigliar.it)
Immacolata Cozzolino (coimma@yahoo.it)
Anna Lucia Peluso (peluso@dbbm.unina.it)
Laura Virginia Sosa Fernandez (laurasosa@hotmail.it)
Claudio Bellevicine (claudiobellevicine@gmail.com)
Giancarlo Troncone (giancarlo.troncone@unina.it)
Marco Picardi (marco.picardi@unina.it)
Fabrizio Pane (fabrizio-pane@unina.it)
Pio Zeppa (pzeppa@unisa.it)

Version: 2 Date: 8 April 2013

Author’s response to reviews:

(1) Ethics Statement: Due to the retrospective, non interventional nature of this study, no consent was requested by the local research ethics committee. This sentence has been merged in the section Material and Methods.

(2) Consent Statement: At the time of FNC, patients were informed of the diagnostic procedure and its related risks; informed consent for the FNC performing, the diagnostic procedures and the scientific use of biological material was obtained from all the patients. This sentence has been merged in the section Material and Methods.

(3) The authors declare that they have no competing interests. This sentence has been added after the section Material and Methods.

(4) Authors’ contributions: PZ, EV: conception and design, interpretation of data, given final approval of the version to be published; IC, ALP, LVSF, CB, GT, MP, FP: acquisition of data, drafting the manuscript, PZ, EV: critical revision. All authors read and approved to be published. This sentence has been added after the section Material and Methods.

(5) Acknowledgments: no acknowledgments have been reported.

(6) Please remove the duplicate figures from the main manuscript.: Figures have not been modified and hence have not been removed.